Author's response to reviews

Title: Susceptibility to intestinal infection and diarrhoea in Zambian adults in relation to HIV status and CD4 count

Authors:

Paul Kelly (guts@coppernet.zm)
Jim Todd (jim.todd@mrcuganda.org)
Sandie Sianongo (guts@coppernet.zm)
James Mwansa (jclmwansa@yahoo.ca)
Henry Sinsungwe (guts@coppernet.zm)
Max Katubulushi (mkatubulushi@yahoo.co.uk)
Michael Farthing (mfarthing@sussex.ac.uk)
Roger Feldman (feldmanmail@aol.com)

Version: 4 Date: 6 January 2009

Author's response to reviews:

We have made the formatting revisions requested, particularly to the Tables.